

Contents lists available at ScienceDirect

# Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser



# Transdisciplinary research promoting clean and resilient energy systems for socially vulnerable communities: A review

Sara Belligoni<sup>a,b,\*\*</sup><sup>(b)</sup>, Elizabeth Trader<sup>a,c,d,\*</sup><sup>(b)</sup>, Mengjie Li<sup>a,d,e</sup>,

Mohammad Siddiqur Rahman<sup>a,f,g</sup>, Javed Ali<sup>a,f,g</sup>, Alejandra Rodriguez Enriquez<sup>a,f,g,h</sup> Meghana Nagaraj<sup>a,f,g</sup>, Sanam K. Aksha<sup>a,i</sup>, Kelly A. Stevens<sup>a,d,e,i</sup>, Thomas Wahl<sup>a,f,g</sup>, Christopher T. Emrich<sup>a,g,i</sup>, Zhihua Qu<sup>a,c,d</sup>, Kristopher O. Davis<sup>a,d,e,j,k,\*</sup>

<sup>a</sup> University of Central Florida, 4000 Central Florida Blvd, Orlando, 32816, Florida, USA

<sup>b</sup> Puerto Rico Research Hub, University of Central Florida, 12815 Scholarship Dr, Orlando, 32816, Florida, USA

<sup>2</sup> Department of Electrical and Computer Engineering, University of Central Florida, 4000 Central Florida Blvd., Orlando, 32816, Florida, USA

<sup>d</sup> Resilient Intelligent Sustainable Energy Systems (RISES) Center, University of Central Florida, 4000 Central Florida Blvd., Orlando, 32816, Florida, USA

<sup>f</sup> Department of Civil, Environmental, and Construction Engineering, University of Central Florida, 4000 Central Florida Blvd., Orlando, 32816, Florida, USA

<sup>g</sup> National Center for Integrated Coastal Research, University of Central Florida, 4000 Central Florida Blvd., Orlando, 32816, Florida, USA

<sup>h</sup> Institute for Environmental Studies (IVM), Vrije Universiteit Amsterdam, De Boelelaan 1111, Amsterdam, 1081, HV, the Netherlands

<sup>1</sup> School of Public Administration, University of Central Florida, P.O. Box 161395, Orlando, 32816-1395, Florida, USA

<sup>j</sup> Department of Materials Science and Engineering, University of Central Florida, 4000 Central Florida Blvd., Orlando, 32816, Florida, USA

<sup>k</sup> CREOL, The College of Optics and Photonics, University of Central Florida, 4000 Central Florida Blvd, Orlando, 32816, Florida, USA

## ARTICLE INFO

Keywords: Climate change Extreme weather Photovoltaics Resilient energy Social vulnerability Solar energy Geospatial Power outages

## ABSTRACT

Extreme weather events caused by climate change can affect the energy sector in different ways. For example, extreme heat, cold spells, strong winds, or flooding may lead to increased energy demand and consumption, reduced energy production, or cause infrastructure failures and outages. Underserved communities are among those most impacted by power outages resulting from extreme weather events due to lower infrastructure investment in the areas where they live. These phenomena encompass a variety of social and technical challenges, for which we propose a new, transdisciplinary framework to explore solutions for providing clean, affordable, and resilient energy systems to vulnerable and at-risk communities. The authors consider a new approach using perspectives from engineering, hazards science, and policy studies to identify and develop solutions for the expansion of the use of solar energy production coupled with increased storage capacities in places where power outages and social vulnerability intersect.

#### Nomenclature

		Abbreviations and Acronyms	Full Term or Phrase
Abbreviations and Acronyms	Full Term or Phrase	DOT	
		BGT	Boosted Gradient Tree
PV	Photovoltaics	BART	Bayesian Additive Regression Tree
LCOE	Levelized cost of energy	ENS	Ensemble
EIA	U.S Energy Information Administration	MedVI	Medical Vulnerability Index
DOE	United States Department of Energy	GIS	Graphic Information System
NASA	National Aeronautics and Space Administration	EMS	Emergency Medical Services
RF	Random Forest	NREL	National Renewable Energy Laboratory
DT	Decision Tree	reV	Renewable Energy Potential
	(continued on next column)		(continued on next page)

(continued)

(continued on next page)

\* Corresponding author. University of Central Florida, 4000 Central Florida Blvd, Orlando, 32816, Florida, USA.

\*\* Corresponding author. University of Central Florida, 4000 Central Florida Blvd, Orlando, 32816, Florida, USA.

\*\*\* Corresponding author. University of Central Florida, 4000 Central Florida Blvd, Orlando, 32816, Florida, USA.

E-mail addresses: sara.belligoni@ucf.edu (S. Belligoni), elizabeth.trader@ucf.edu (E. Trader), kristopher.davis@ucf.edu (K.O. Davis).

## https://doi.org/10.1016/j.rser.2025.115434

Received 8 September 2023; Received in revised form 26 December 2024; Accepted 23 January 2025 Available online 9 February 2025

1364-0321/© 2025 Elsevier Ltd. All rights are reserved, including those for text and data mining, AI training, and similar technologies.

<sup>&</sup>lt;sup>e</sup> Florida Solar Energy Center, University of Central Florida, 1679 Clearlake Rd., Cocoa, 32922, Florida, USA

S. Belligoni et al.

(continued)

Abbreviations and Acronyms	Full Term or Phrase
SAM	System Advisor Model
ITC	Investment Tax Credit
EPA	United States Environmental Protection Agency
EO	Executive Order
kWh	kilowatt-hour
MW	Megawatt

## 1. Introduction

Climate change is increasing the frequency and magnitude of extreme weather events worldwide. In particular, there are more extreme temperatures, heavy precipitation, and worsening droughts in some regions [1]. This phenomenon is having strong effects on communities, especially those that may be more exposed to extreme weather events, either because they are more prone to experience natural hazards or because their socioeconomic characteristics make these communities more vulnerable. Numerous other studies are identifying several patterns that connect impacts from extreme weather events and socioeconomic factors [2].

Extreme weather events are being captured by macroeconomic indicators, suggesting that they are starting to have an impact on both economic growth and national gross domestic product in several countries. Fig. 1 shows the economic impact of weather events and climate disasters in the United States (U.S.) using data from the National Oceanic and Atmospheric Administration [3–5]. Tropical cyclones, severe storms, and drought have had the greatest economic toll, but all of the disaster categories have led to significant loss. In addition to the economic impacts, the increase in global temperature is also affecting the agricultural sector, as crop production has been threatened by extensive droughts. The impact on agriculture, as well as other sectors, can have a domino effect on labor dynamics as several people find themselves with reduced sources of income, making some people more socioeconomically vulnerable as a result of climate-induced extreme weather events [2].

Extreme weather events affect the energy sector in different ways, ranging from increased energy demand and consumption to reduced energy production, infrastructure failure, and grid outages. Each of these is projected to change in frequency and severity in North America given our warming world [6]. Some notable recent examples that show the significant human and economic tolls related to grid outages were Hurricane Maria in 2017 in Puerto Rico [7], Hurricane Michael in 2018 in the Florida panhandle [8], and the Texas power crisis in 2021 due to extreme cold [9]. Oftentimes, vulnerable or marginalized segments of society are most affected by these extreme events due to social inequities and are already struggling to pay their energy bills [10–12].

As extreme weather events become more common and electricity use continues to grow, identifying which areas are most vulnerable to the negative effects of these events becomes crucial to increasing resilience [13]. Resilience represents the ability of a system to absorb and adapt to human-induced or environmental disturbances, ideally by recovering to a stronger condition [14–16]. The approach to developing solutions to these resilience challenges has primarily been focused on the technical elements within engineering [17]. Fig. 2(a) shows the total number of research articles related to each U.S. state for all types of hazards, while Fig. 2(b) shows the distribution of research articles per hazard type for all states.<sup>1</sup> The scholarship efforts regarding power outages for each state do not necessarily mirror the actual economic impacts, shown in 2

(c) for all types of disasters and hazards and 2(d) for tropical cyclones only.<sup>2</sup> When breaking down the distribution of research across various hazard types within each state, it is surprisingly similar, demonstrating a redundancy of existing approaches towards studying this topic. As the authors propose in this review, a more transdisciplinary approach is needed to evaluate the complex relationships between the social and physical sciences to address grid stability in socially vulnerable areas.

One approach to solving this problem lies in expanding the use of photovoltaic (PV) energy production coupled with energy storage in places where power outages and social vulnerability intersect. Not only would this increase energy resilience, especially during extreme weather events, but also has the potential to lower electricity costs for lowcapacity communities [20] and reduce air pollution which has been linked to negative health impacts [21]. It can also provide households with the energy required to keep communication and medical devices charged and to provide enough cooling to prevent related medical episodes, such as during Hurricane Irma (2017) where the most common indirect cause of death was exasperation of an existing health issue [22]. However, various challenges remain that need to be addressed to bring forth solutions to improve electric grid reliability and provide clean, affordable, and resilient energy to socially vulnerable communities.

Capital is finite, and PV systems with energy storage (PV + storage) are capital-intensive investments. Often, the communities most vulnerable to grid outages caused by extreme weather events lack the financial resources to build out PV + storage infrastructure [20]. Policies should be put in place to encourage investment in these resources in vulnerable communities. The most effective policies should more precisely identify the locations most at risk in a physical and social sense. Additionally, the most appropriate locations for PV + storage are also dependent on technological constraints related to the electric power system (i.e., the grid). Increasingly, research on resilient energy systems such as microgrids identifies the need for more participatory planning with an actively engaged population for where resilient systems will be used [23]. Fig. 2 (e) and (f) include the energy expenditures per capita and estimated potential annual energy production available for rooftop PV in each state. Collectively, this provides an interstate comparison of the financial burden associated with energy consumption and the energy potential for rooftop PV.

Another challenge is related to energy equity, which is the concern for fair and just access to energy services, resources, and benefits across all populations. For example, solar panels provide clean, low-cost electricity, however are more commonly adopted by higher-income households [24], indicating there are still significant energy equity implications to consider during this transition. An equitable approach takes into consideration the different experiences across populations while also actively engaging all impacted communities in the decisionand policy-making process [25]. Energy equity issues stem from a variety of sources, such as high energy prices in low-income areas, or pollution from energy systems inflicted on communities with fewer economic and political resources. Households that are not able to afford their electricity bills face higher rates of energy burdens which may lead to adverse health effects from compromised living standards [26]. Access to clean, distributed energy is one way to significantly improve health outcomes and energy resilience for underrepresented areas [27]; however, careful consideration of program and policy design choices is needed to maximize equity [24]. Similarly, the concept of "equitable resilience" considers unequal access to resilience that goes beyond social vulnerability to also incorporate access to resources, participation into the decision- and policy-making arena, and ability to take part in resilience planning and development [28]. Deployment of more resilient energy solutions, such as PV + storage, requires considerable financial investment, therefore careful study and planning is needed to place

<sup>&</sup>lt;sup>2</sup> Tropical cyclones are highlighted here because they have caused the most economic loss and are capable of causing widespread power outages.



Fig. 1. Cost in millions of U.S. dollars of weather events and climate disasters with impacts over one billion U.S. dollars shown as: (a) the total cost per disaster type; and (b) the cost per year for each disaster type. The data source is [3].

these assets where they can benefit communities equitably [20].

In this review, the case is made for a more transdisciplinary approach to evaluating the challenges and need for resilient energy systems that include multiple perspectives including engineering, computer science, hazards geography, technoeconomic analysis, and public policy. This article reviews relevant work from these various disciplines and proposes a transdisciplinary, geospatial framework focused on equitable grid resilience. Transdisciplinarity refers to an approach that places the interactions among several research disciplines in a holistic and integrated system that results in an evolving and collaborative subdiscipline [29]. Fig. 3 illustrates the key elements of this proposed framework and how each element fits together. This review is organized in the following manner. Section 2 reviews the availability of relevant data sources, analysis and modeling methods, and prior research findings related to connecting extreme weather to power outages in a spatially-resolved manner. This part of the overarching framework, shown on the bottom left of Fig. 3, relies on the use of historical outage and weather data to build models that can predict risk of outages for geographic regions of interest while considering a changing climate. Section 3 covers the links between social and medical vulnerability and power outages. Socioeconomic an medical data can be used to build models for quantifying vulnerability (bottom right of Fig. 3). This can be combined with risk maps to identify geographic regions and, depending on the spatial granularity of the data and models, specific communities that are both at an elevated risk of experiencing outages and socially or medically vulnerable. Section 4 discusses the interventions and technological solutions available to increase energy resilience with an emphasis on the PV and energy storage systems (PV + storage). And finally, section 5 covers the policy implications and potential policy solutions. In the proposed framework, these elements would all be brought together, along with stakeholder feedback, and then used to determine the most appropriate technology and policy solutions in a spatially-resolved, and therefore community-specific manner. There are challenges in working with massive, heterogeneous datasets, but approaches have been developed using high-performance computing and distributed storage [30].

#### 2. Extreme weather and power outages

Extreme weather events have become a major cause of electric power outages in recent years. In the U.S., the annual impact of weather-related blackouts (i.e., power outages specifically) ranges from \$20 to \$55 billion in U.S. dollars [31]. In the U.S., an average of 520 million customers were affected by power outages across 2447 counties between 2018 and 2020; outage data show that about 62 % percent of the outages

lasted 8 h or more and were caused by hurricanes, extreme storms, and heat [10]. High winds during hurricanes and storms are considered one of the main causes of wide-area electrical disturbances. For example, Hurricane Sandy (2012) resulted in 8.5 million people being without power, and Hurricane Michael (2018) resulted in 1.7 million people being without power [32,33]. In Florida, hurricane-induced flooding and wind, are the two most frequent causes of power outages [34].

In addition to the wind, other weather drivers result in direct damage to the electricity system infrastructure. Lightning strikes that affectoverhead conductors can result in short-circuit faults, which in turn leads to electrical protection and disconnected lines. Cold waves, heavy snow, and ice accumulation increase the likelihood that overhead lines and towers fail. Under freezing conditions, ice and snow may gather on insulators bridging the insulators and provide a conducting path, resulting in flashover faults. In the winter of 2021, Texas suffered three major winter storms that produced equipment problems (i.e., power equipment in Texas was not winterized) and high demand for electricity. At the peak of the crisis, 4,011,776 customers experienced power outages throughout Texas [35]. Wildfires also impact the power grid, a recent example is the 2021 Colorado wildfire, also known as Marshall Fire, which impacted Xcel Energy's natural gas infrastructure that supports the system in two Counties. High temperatures and heat waves limit the transfer capability of transmission lines and increase the energy losses and the line sagging.

The frequency of power outages caused by extreme weather events is already on the rise, driven by the increasing occurrence of severe weather events and the aging infrastructure of the power grid. 2017 is to date recorded as the costliest Atlantic Hurricane Season since 1851, the year in which losses and damages assessments began to be recorded. 2017 was followed by the 2012 and 2008 seasons which also recorded several weather-related power outages [36]. The increase in consumer power outages, partially due to an increase in extreme weather events [37], also correlate with increases in related health incidents such as increases in carbon monoxide poisoning from generators [38]. As a result, it is important to identify areas that are most vulnerable to power outages due to extreme weather events.

## 2.1. Power outage data sources

In reviewing existing literature pertaining to this issue, there is a dearth of spatially granular data related to power outages. While community aggregated data has been used to reveal disparate impacts during this extreme events such as the Texas winter storms [39–41], there are limited publicly available sources of large-scale power outage data with the type of spatial resolution needed to locate specific communities



**Fig. 2.** The number of research articles published related to extreme weather and power outages shown: (a) for all hazards as a map of U.S. states all hazards; and (b) for all states, but broken down based on hazard type. Maps of U.S. states showing: (c) the total cost in millions of U.S. dollars of large weather events and climate disasters since 1980 for each state from Ref. [3]; (d) cost in millions for tropical cyclones specifically; (e) annual energy expenditures per capita [18]; and and (e) estimated potential annual production from rooftop PV per capita estimated from Google's Project Sunroof [19].

that have disproportionately experienced outages. This highlights the need for a replicable methods of extracting power outage data on a granular level and public sources of this data.

Specifically within the U.S., many of the public outage datasets are reported at the county of city level, like data from the U.S. Department of Energy Form OE-417 submission [34] and the more recent EAGLE-I dataset [42,43]. However, in areas that are geographically large, have high populations, and are socioeconomically diverse such as Miami-Dade County in Florida, county or city-wide averages may mask more vulnerable populations at the neighborhood level.

As for private direct access to power outage data, electric utility companies are the primary source of outage data in their transmission area. However, utility companies are disincentivized to share this information if they feel that it may create financial, legal, or public relations liabilities. Another method of accessing the power outage data directly is through smart meter data provided by consumers (e.g. Pecan Street). However, that requires large consumer cooperation and mobilization in order for it to be scalable across many areas.

In the absence of direct power outage data from electric utilities, one option is to process information for data sets whose contents correlate with extreme weather resilience. For example, National Aeronautics and Space Administration (NASA) nighttime satellite imagery can be used for the identification of power outages based on the visibility of lights. This is an indirect method of measuring power outages, however, it can strongly correlate with power outages during extreme weather events. For example, the nighttime images taken via NASA satellite before and after Hurricane Michael in 2018 as depicted in Fig. 4 show the extensive power outages in the area as a result of the storm [44]. There is a clear



Fig. 3. Illustration of the proposed transdisciplinary, geospatial framework for promoting clean, affordable, and resilient energy to socially vulnerability communities.



Fig. 4. NASA nighttime satellite imagery (a) before and (b) after Hurricane Michael, on October 6, 2018 and October 12, 2018, respectively. Image sourced from Ref. [44].

abundance of areas that experienced a decrease in measurable light.

The limitation of using indirect data sets such as these is that there is a degree of noise that has to be filtered out. In this case, the type and abundance of cloud coverage affects the amount of light shown, therefore another resource is needed to verify cloud coverage at any given pixel during the time that this was taken in order to improve the accuracy. Additionally, other indirect sources for power outage data are not always accessible. For example, it may be possible to conduct a social medias' posts analysis by identifying keywords that may find posts discussing power outages in a given area. This could be possible by making use of tools for social media content analysis (including embed social medias resources such as those available in Facebook) and apply them to individuals and groups (such as, neighborhood Facebook groups) posts. Additionally, another potential source of indirect data is food spoilage insurance claims, however obtaining those claims would involve requesting this information from the company that fulfilled the claims, who may not be willing to cooperate.

## 2.2. Parameters that affect power outage

Power outages often occur due to the complex interaction between various weather variables, vegetation cover, and infrastructure. Most weather-related power outages are caused by falling trees or their limbs affecting electricity distribution lines and poles. Major outages are more often caused by damage to electricity transmission lines [31]. Thus, vegetation management is crucial to reduce the risk of power outages. This is well reflected in the performance of the weather-outage models; they have shown lower accuracy if vegetation information (usually tree trimming) is not accounted for. The electric system infrastructure (e.g., electric transformers, fuses, reclosers, switches), the number of customers served, and the type of soil have also been found crucial to model power outages. Although the weather-only model has shown lower accuracy than models including infrastructure and vegetation information, they still perform fairly well [45–48].

The relative relevance of the weather variables in the outageweather model depends on the region [49] and on the type of weather event. In Connecticut [45], concluded that total precipitation, maximum and mean temperature, and mean wind gusts are the most important variables in extra-tropical cyclones while maximum specific humidity, maximum and mean temperature, and maximum soil moisture are the most significant weather variables in thunderstorms. Importantly, models which ignored near-surface parameters were unable to adequately predict outages related to thunderstorms. Staid et al. [50] used tropical cyclone winds to obtain the number of customers without power for each census tract in the Gulf of Mexico and the Atlantic coast of the U.S.

Usually, the set of weather variables also includes precipitation and pre-storm condition indices such as soil moisture at different depth levels [45,47,48], temperature [45,51,52], humidity [45,51], drought [53] and storm intensity classification [45,51], as well as the total number of customers served by the utility in a grid cell and information about tree trimming. It was shown that when the tree-trimming variable is removed from the model, the performance drops, but the model can still predict outages with reasonable accuracy. Finally, results suggested that models which only consider gust wind speed, as is typical for engineering fragility curves, are unlikely to accurately model failures in the power system.

Despite the extensive literature on storm-weather power outage models, some hazards have been widely neglected such as flooding from surge and/or precipitation. Personal communication with utility companies highlighted that flooding is particularly relevant for power outages, especially in hurricaneprone areas [54]. used a hurricane outage model to predict customer outages for Hurricane Sandy in the Northeast of the United States. Although the model did well for many regions, it underestimated customer outages in Connecticut. It was hypothesized that this may have been because storm surge was not included in the model. However [47], found that the storm surge only contributed to a minor fraction of the customer outages in the region while the majority was related to trees interacting with the overhead lines. The existing discrepancy regarding the relative importance of flooding in power outages underscores the need for more comprehensive assessments on the topic.

The above-mentioned weather events cause direct damage to the power system infrastructure, but weather events can also have indirect effects. Sustained heat waves often cause significant changes in the load, comlicating electricity demand predictions. Unanticipated higher demand for space conditioning and refrigeration during heat waves has led to rolling outages in the past. During a heat wave in September 2022, a California grid operator issued a power grid emergency alert to conserve electricity use to prevent outages. In this context [55], found that the cooler demand could be underestimated if only air temperature is accounted for, neglecting the effect of humidity, as it is usually done. They also showed that the relative importance of humidity and temperature varies greatly across a region.

# 2.3. Machine learning models

In order to better anticipate and reduce the impacts of weatherinduced power outages, several studies have developed forecast models to derive the extent and magnitude of the outages. A summary of these is provided in Table 1. Usually, models are designed to predict the expected number of outages, defined either by the number of customers Table 1

Summary of machine learning approaches to build weather-outage models.

Target	Input	Model(s) Tested	Best Model
Number of outages per 2 km grid cell	Weather, grid infrastructure, land cover, vegetation [45]	Decision tree, gradient-boosted tree, random forest, ensemble regression, and BART	BART
Count of outages per grid cell (>5 min) [47]	Simulated weather for 89 storms using WRF model variables including wind gust, wind stress, wind at 10 m height, soil moisture, precipitation rate joined with grid information, land cover information	Decision tree, gradient-boosted tree, random forest, and ensemble regression	Ensemble regression
Outage duration caused by hurricane [48]	Land use and land cover, hurricane duration and intenstiy, precipitation, soil moisture, wind speed	BART, regression trees, accelerated failure time (AFT) and Cox proportional hazard models (Cox PH)	BART
Electricity consumption (GWh/month) [56]	Temperature, precipitation, wind speed, wind gust, visiblity, dew point temperature, and economic data	BART, GLM, GAM, MARS	BART
Outage duration [57]	Wind gust, precipitation, ice accretion, land cover, duration of strong winds	AFT	AFT
Number of outages [58]	Hurricane characteristics, land cover, non-hurricane climatic data, grid information	GLM, GAM	GAM
Major outages at the state level [52]	Socioeconomic data, electricity consumption, state- level climate and weather data, land use data, grid information	SVM, RF	Twostage hybrid risk estimation model
Utility-reported power outages for each storm event [59]	Weather data from ERA5 and ERA5-Land, grid information, land cover, tree canopy cover	GBM, RF, optimized model (OPT)	OPT

without power or the number of locations that require manual intervention to restore power. A diverse range of machine learning techniques exists, and they have been applied individually or in conjunction with each other in numerous ways. The most frequent machine learning algorithms used include random forest (RF), decision tree (DT), boosted gradient tree (BGT), an ensemble of models of RF and DT (although ensembles of other models can be found in the literature), and Bayesian additive regression tree (BART) [45,47,48,56]. To a lesser extent, statistical models such as the generalized linear model and the generalized additive model have been also applied to predict power outages [57,58]. Overall, weather-outage models use the same type of input predictors/parameters, which includes information on infrastructure (i.e., a precondition of the power grid), vegetation (e.g., cover, type of vegetation, and trimming), and weather conditions (e.g., wind, precipitation, temperature).

The literature describes a best practice for using machine learning techniques in analyzing impacts on the electric power system. Work from Ref. [47] compared RF, DT, BGT, and an ensemble (ENS) of the three models to predict the number of outages per grid cell in

Connecticut, U.S. for 89 storms including thunderstorms, blizzards, Nor'easters, and hurricanes. Combining magnitude and spatial accuracy, the ENS performed the best, followed by the RF [45]. compared the same set of models (DT, BGT, RF, ENS) in addition to BART, for 76 extra-tropical storms and 44 convective storms in Connecticut. The BART model performed better than others, followed by BGT, ENS, DT, and RF. A similar conclusion was drawn by Ref. [48], who predicted the mean outage duration in the central region of the Gulf of Mexico using data for three different hurricanes. They tested the accelerated failure time regression, cox proportional hazards, BART, and a multivariate adaptive regression spline. They found that the BART model outperformed all other statistical models [56]. tested five different models (generalized linear model, generalized additive model, multivariate adaptive regression splines, ENS-based tree model, and BART) to predict power consumption in Florida, U.S. and found BART to outperform other models. Furthermore [52], developed a two-stage hybrid machine learning model combining support vector machine and RF to predict the extreme outages in the USA at the state level [59]. proposed a framework to classify the severity of an outage event and prediction based on RF, BGT, and ENS (RF + GBT) models for the Connecticut service territory of Eversource Energy. They found that the ENS showed better performance in comparison to single models. Overall, based on the above-mentioned studies, the BART model has been found to yield better outage predictions, followed by ENS.

# 2.4. Climate change impact on power outages: relevance in Florida

Climate change is expected to affect many of the weather variables discussed above and hence will likely also affect the operation and reliability of power systems [60]. For example, there are temperature thresholds above which the operation of transformers and overhead lines is impacted. With the expected continuous increase in the ambient temperature certain elements of the system will likely have to be derated. The same changes in temperature will likely also negatively affect the efficiency of thermal power plants [60]. The continued rise in sea level will threaten coastal assets such as oil and gas pipelines and also facilitate coastal flooding due to elevated base water levels onto which tides and storm surges are superimposed. Changes in rainfall patterns will lead to more flooding in certain areas and more drought periods in others. The former can directly impact power system infrastructure, while the latter affects hydropower generation and the availability of water for cooling purposes in thermal and nuclear power plants [60].

The U.S. Department of Energy in its Regional Vulnerabilities and Resilience Solutions report (2015) [61] indicated that the Southeast region of the country hosts a large amount of energy infrastructure in low-lying coastal plains that are vulnerable to increases in flooding. High winds, coastal erosion, flooding, and large waves from hurricanes and sea level rise-enhanced storm surges threaten oil and gas production, ports, pipelines, refineries, and storage facilities, as well as electricity generation and transmission assets. Higher temperatures and more frequent, severe, and longer-lasting heat waves are also projected for the Southeast, potentially increasing peak electricity demand for cooling while reducing the capacity of the thermoelectric generation and transmission systems needed to meet the increased demand. In this regard, Staid et al. [50] identified the Houston, Texas, New Orleans, Louisiana, and Miami, Florida, metropolitan areas to be heavily impacted even for scenarios of lower-intensity storms. On the contrary, Alemazkoor et al. [62] concluded that hurricane-induced power outage risk under climate change is predominantly driven by the uncertainty in the future frequency of hurricanes. Further research is needed to assess changes in the frequency of extreme events taking into account the uncertainties of projections.

Florida is not exempt from the extreme weather events experienced in the Southeast region. The state encounters various climatic phenomena, heat waves, inland/coastal floods, and strong winds, the latter two often resulting from hurricanes. In addition, the state is experiencing a steep increase in population, which can burden the power system under certain weather conditions such as high temperatures. As mentioned above, the frequency and magnitude of these extreme weather events are expected to change under a warming climate. There have been strong developments in short-term power-outage prediction models, however, the evaluation of the changes in long-term power outages has been limited to the study of hurricane frequency and uncertainties [62] and for changes in the cooling demand [55]. There is a need for infrastructure providers and emergency managers to plan on much longer time scales as recognized by other authors [50], including not only temperature and hurricanes but all other weather parameters that have been recognized to influence power outages.

## 2.5. Local study areas vs. large study areas

As mentioned above, power outages are a function of weather conditions, vegetation cover, and infrastructure, all of which exhibit significant spatial variation. In addition, the impacts of climate change on weather patterns also vary across different regions. Therefore, conducting local studies becomes imperative in order to capture the characteristics of specific areas and deliver more accurate short-term power outage forecasts, as well as long-term projections. By utilizing localized weather-outage models, it becomes possible to forecast weather events in a more realistic manner and gauge the resilience of specific regions against fluctuating weather conditions.

## 3. Social and medical vulnerability

Understanding the needs of vulnerable people is of utmost importance, especially among emergency managers, public health officials, social workers, and social service professionals. However, in cases of disasters or emergencies where sensitive populations may require aid in adequately preparing for, responding to, recovering from, or mitigating hazards, emergency management officials and planners need detailed information to anticipate and meet often very specific needs [63]. From this perspective, vulnerability represents the potential for loss or harm among individuals and communities facing power outages stemming from hazards, disasters, or energy system insecurity, such as those we observed in the Texas 2021 winter power failure [64,65] or those that occur every time a hurricane makes landfall. In hazard literature, the vulnerability concept broadly includes the structural vulnerability of the infrastructure system (power system), often called a biophysical vulnerability, and uneven exposure of individuals and households (to the power system) based on social, political, and economic factors, usually expressed as social vulnerability [66,67]. Generally, social vulnerability refers to those characteristics of a person or group and their situation that influence "their capacity to anticipate, cope with, resist, and recover from the impact of a natural hazard (an extreme natural event or process)" [68]. From a practical perspective, understanding social vulnerability determinants means that decision-makers should be able to determine what supplies, equipment, and personnel are needed to respond effectively in emergencies which requires sound knowledge of a region's social, economic, and baseline health conditions.

Consequently, understanding the vulnerability of a population requires a location-based assessment of both socioeconomic sensitivities and special medical needs, which may represent different subpopulations. Although various methods are employed to empirically assess those needs to understand the spatial distribution of vulnerable and marginalized populations, the social vulnerability index (SoVI) is the most popular and widely used in hazard planning, disaster research, and decision-making in emergency management [67]. SoVI models the geographic distribution of vulnerable populations based on 29 different variables that stem principally from vulnerability literature: employment structure, housing, population structure, race/ethnicity, socioeconomic status, and special needs [63]. Using a statistical data reduction approach, these variables are reduced to major components of vulnerability, which can later be mapped using geospatial tools and techniques.

Additionally, supporting the medical and health needs of impacted populations is another challenge to emergency managers during disasters and emergencies. An empirical assessment method called Medical Vulnerability Index (MedVI) is used to understand the medical necessity of an area of interest [69]. MedVI is based on the theoretical framework of social vulnerability and is a companion concept of SoVI utilizing a similar methodological approach. MedVI is constructed from two key concepts; health needs and healthcare access, from medical vulnerability literature and is represented by 36 indicators in four distinct themes: physical health needs, psychological health needs, social health needs, and healthcare access [70,71]. The variables are reduced to major principal components of medical vulnerability using factor reduction technique and have the utility of mapping using a Geographic Information System (GIS) environment.

# 3.1. Social and medical vulnerability and power system failure

Advanced industrial societies are becoming more technologydependent and are thus more vulnerable to technology failures. Power systems, one of the critical infrastructures of modern societies, are exposed to various activities such as severe weather, intentional attacks, equipment failures, fuel supply emergency, and islanding [38]. Power system failure is identified as "the single most vulnerable system in our critical infrastructure" [72], and such disruptions and failures will become more frequent with increasing climate extremes and affect more people [38]. Power system failures do not impact individuals equally, and access to proper resources (or lack thereof) can significantly affect how individuals deal with long-duration outages [73]. The relationship between social vulnerability and power outages has recently been explored [10,73–75]. This nascent literature shows that various socioeconomic and demographic characteristics play significant roles in increasing health risks, power outage preparedness, and willingness and means to evacuate if necessary [76-80]. These early results highlight a need to identify socially vulnerable groups and medically vulnerable communities so that adequate resources, information, and assistance can be prepared in a more targeted way during such events.

Past studies have shown that differential exposure to such failure dramatically impacts sub-populations. For example, the collapse of the power grid system in 2003 in North America caused ninety excess deaths in New York City, New York, alone, a rise in mortality of 28 %. The death rates were highest for those aged between sixty-five and seventy-four, which shows the differential sensitivity of the population to such events [77]. A study in the Midwest region of the United States indicates that ZIP codes with lower median household income have experienced more power outages during the study period between 2017 and 2022 [81]. Other research found that American Indian populations were positively correlated with average outage duration at the census block group level, showing unequal exposure to such failures [11]. Additionally, the existing literature suggests that power outages have significant health consequences ranging from carbon monoxide poisoning, temperature-related illness, gastrointestinal illness, and cardiovascular, respiratory, and renal disease hospitalizations, especially for individuals relying on electricity-dependent medical equipment [38]. After a 2003 power system failure, hospital emergency department visits and admissions due to respiratory problems increased, especially among women, elderly persons, and people suffering from chronic bronchitis [76]. During a derecho event (destructive thunderstorms) in West Virginia in 2012, utilization of Emergency Medical Services (EMS) and hospital resources increased substantially, from EMS scene responses to inter-facility transfers and standbys increasing by more than 50 percent over the previous year [82]. Thus, power system failure studies should consider factors such as socioeconomic and other social and medical

vulnerabilities as well as how community resiliency can mitigate and minimize the adverse impacts of widespread major power outages.

# 4. Technological solutions

The provision of reliable and affordable electricity is essential for economic growth and people's lives. The best way to achieve power grid resilience is through holistic resilience planning [83], as well as continuous advancement of grid technology and diversification of sustainable energy resources.

Technological solutions to achieve resilience proposed in existing literature often require some or all of the following.

- Installation and diversification of additional energy resources;
- Increase of power flow capacity, and reconfigurability of a power system into multiple microgrids [84];
- Optimization, control, and automation to increase flexibility, redundancy, decentralization, self-organization, and coordination [85];
- Implementation through transparency, collaboration, and foresight considerations [85].

Also, every power system has unique features, and hence solutions must be developed for specific communities and their circumstances. And, a combination of tailored solutions should be used to address all potential vulnerabilities.

To achieve resilience of the main grid under extreme weather conditions existing literature suggests the necessity for major generation assets as well as transmission networks need further diversification and decentralization. Examples of these include: (1) hardening grid infrastructure, including transmission line fortification, flood prevention of substations, and transmission capacity enhancement [85]; (2) diversification of generation mix to sustain power generation under supply disruptions and extreme weather conditions; (3) spatial distribution and decentralization of power generation by adding utility-scale wind farms, PV farms, energy storage, and other assets (such as hydrogen); and (4) enhance interconnections among regional grids. At the distribution level, resilience enhancement measures include: (1) underground distribution lines; (2) islandable and interconnected microgrids for local infrastructure needs and critical loads; (3) community PV and energy storage to power communityshared facilities such as a resilience hub; and (4) behind the meter distributed generation, such as rooftop solar and energy storage, coupled with grid-forming inverters (which establish the voltage to supply electricity to individual homes during grid outages) and smart home electricity panels (which enable classification and grouping of various loads in the homes and make them controllable loads). Upon choosing a specific resilience strategy, additional studies should be undertaken to ensure operational reliability, effectiveness, and economic benefits.

From the technical perspective, renewable energy resources such as PV and wind are highly variable, and there is a limit on how many variable resources can be incorporated into a distribution network without violating operational voltage limits. Determination of the limit, referred to as hosting capacity analysis [86], can be done using power system simulations. Typically, distribution networks would experience voltage issues when the renewable penetration level exceeds 20 %. There are several ways to increase the penetration level and maintain a stable and admissible voltage profile across the distribution network.

The first is the incorporation of battery-based energy storage to smooth out the PV production variability (i.e., PV + storage), often called PV smoothing, and the trade-off is the cost. The others are cooperative controls of both reactive power and active power (the latter of which can be accomplished by either storage control or load control or both). To achieve a high level of resilience, it is desirable to raise the PV penetration level to the maximum base load, i.e., at or over 100 % penetration so that the excess PV generation during the PV peak hours

can charge the storage devices within the network. To enable such extreme penetration, the so-called control enable hosting capacity analysis should be carried out, and a distributed cooperative control needs to be implemented to coordinate the control actions among the distributed energy resources. Both of them are embedded in the open-source software package MA-OpenDSS [87].

The control-enable hosting capacity determination is a two-stage greedy search algorithm to find realistic worst-case scenarios (in terms of operational voltages) for 100 %-200 % PV penetration, considering both the location and the size of the PVs in the primary and the secondary nodes of a feeder [88]. In the first stage, the worst-case scenario is found by considering PV locations only in the primary nodes. In the second stage, the worst-case PVs from the parent primary nodes are redistributed further down throughout the secondary children's nodes by taking into consideration the size of the PVs. The PVs distributed in the secondary nodes are assumed to not exceed twice the amount of the total load in the secondary node, and any remaining PVs are left on the parent primary node. The process is repeated feeder by the feeder, and the worst-case scenarios for all the feeders are combined to obtain the worst-case scenario for each system. This two-stage greedy search approach reduces the search computational time which, otherwise, would have been intractable for n! combinations required for a feeder with n nodes. Once the worst-case scenarios are identified, the second algorithm of distributed cooperative controls can be applied to perform reactive power compensation so all the feeders in the distribution network can be operated adaptively and autonomously to ensure all voltages are within the operational range (0.95–1.05 per unit).

In Fig. 5, the first sub-figure shows a sample 100k-node system for which the greedy search algorithm determines the worst PV distribution of 100 % penetration. The distribution network voltage profile of the worst-case PV penetration without and with cooperative distributed reactive power control are shown as the second and third sub-figures, respectively. Upon completing control-enabled hosting capacity analysis, the planner can choose multiple configurations of PV and energy storage to perform techno-economic analysis and in turn, make decisions.

A holistic resilience plan and the use of modern grid technologies, like those described previously, enable the integration of PV systems with and without storage. PV systems are inherently clean (i.e., no direct emissions), not subject to fuel supply issues and price volatility, and provide numerous environmental and public health benefits compared to fossil fuel resources [89–91]. Like virtually any technology, the adoption of PV is strongly linked to its economic competitiveness compared to competing options. Technoeconomic analysis plays an important role in evaluating PV systems and PV + storage systems in comparison to fossil fuels, as well as other alternative sources of energy. Often, the levelized cost of energy (LCOE) is one of the key metrics being calculated. The usefulness of LCOE is that it provides a means of comparing different energy generation resources against each other and against, in this case, typical retail electricity rates.

A variety of tools and methods exist for this type of analysis, and they are often dependent on a wide variety of factors. LCOE itself can be simply described as the lifetime cost of the energy system divided by the total energy produced over the life of the system. For a PV or PV + storage system, the majority of the cost is paid at the beginning of the system's life in the form of hardware costs and soft costs (e.g., design, installation, permitting, financing, sales, overhead). The typical warrantied lifetime for PV modules and many of the other components is in the range of 25–30 years normally, so there are naturally some operations & maintenance costs incurred, but these are relatively low for most PV systems.

Ground-mounted utility scale PV systems without energy storage have the lowest LCOE due to lower hardware and installation costs driven by economies of scale and have the largest market share [92], compared to residential and commercial rooftop. The total installation cost per watt (W) for utility scale PV are around \$1.3/W in the U.S. [92], compared to around \$2/W for larger non-residential PV and closer to \$4/W for residential PV [93] depending on the state. However, utility scale PV does not provide distributed resilient power to communities in the case of an outage because it is reliant on transmission and distribution infrastructure. Nor does residential PV if some type of energy storage option, like batteries, is not included. Grid-connected inverters without storage shut down when power from the utility is lost, in order to comply with interconnection standards aimed at preventing unintentional islanding [94-96]. There are various ways to assess the value of backup services to residents, such as using estimates of current electricity rates or what residents are willing to pay to keep their power on. While the average U.S. residential price of electricity is approximate \$0.14/kWh, studies have found that residents are willing to pay between \$0.3/kWh to \$1.2/kWh for backup services to maintain full or partial electric services during a blackout [97]. Achieving LCOE values below \$0.14/kWh is already achievable in the U.S. and abroad [98], even for residential PV + storage systems [99], although these system will always have a higher LCOE than utility scale systems at less than \$0.04/kWh currently in the U.S [92].

In addition to costs, the energy yield of the PV over the life of the system is also critical being the denominator in the LCOE equation. Yield is driven be a variety of factors, including location's expected incident solar irradiance over time (i.e., solar irradiation) and local climate (e.g., ambient temperature), the specific mounting configuration and resulting microclimate, and the specific PV cell and module technology used. Locations with high irradiation are critical, because the output power and therefore yield are directly proportional to irradiance and irradiation, respectively. The power decreases slightly as the PV cell and module operating temperature increase, and the resulting temperature coefficient is strongly dependent on the PV cell technology and module design. The local climate, microclimate, and module selected will also impact the performance loss rate (i.e., degradation rate) of the system



Fig. 5. An 100k-node system: 12 feeders, 53,011 buses, 124 + j41 MVA total load.

#### [100].

A variety of reliability and durability issues have been studied in fielded PV systems and many resolved over the years for older technologies [101-111], but questions remain for newer technologies [112–123]. Additionally, component failures require additional human intervention and add cost. And finally, extreme weather itself can also have an impact on PV system performance [124], although snow loading [125] and hail [126,127] appear to have a larger impact on PV system performance than things like tropical cyclones. Manufacturing metrology and field inspection techniques and analysis methods have been developed to monitor PV performance and identify defect and problems related to manufacturing issues, degradation processes, or extreme weather events [128-139]. A variety of modeling and simulation tools have been developed by the PV R&D community to calculate the expected performance of PV systems using these various inputs, including tools using physical and analytical models (e.g., pvlib [140–142], System Advisor Model [143], PySAM [144], PVsyst [145]) and newer data-driven models [146].

In terms of existing rooftop PV system installations in the continental

U.S., California is the leader in the total number of installations. In terms of the annual potential energy production from rooftop PV, estimates from tools like Google's Project Sunroof indicate virtually all U.S. states have significant potential [19]. Open tools and databases like Project Sunroof, DeepSolar [147], and the U.S. Large-Scale Solar Photovoltaic Database [148] enable people to locate existing PV systems and identify suitable locations for adding PV, a shown in Fig. 6. These tools, when used in conjunction with the grid planning tools described previously, can assist in efforts to promote clean, affordable, and resilient energy systems.

Ongoing research aims to develop innovative machine learning and optimization algorithms that address outage forecast for specific communities, their technology solution of PV + storage, resilient enhancement of distribution network, and their technoeconomic analysis and optimization. While the algorithms must be generic, scalable, and computationally efficient, the technology solution and its corresponding technoeconomic analysis are community dependent because physical location, power system topology, physical attributes of the distribution network, and load/generation distribution all play their roles. In



Fig. 6. (a) Example of Google's Project Sunroof highlighting suitable rooftops for PV in Florida, Orange County, and a specific building in downtown Orlando [19]. (b) Example of the U.S. Large-Scale Solar Photovoltaic Database showing Florida and a specific large, utility-scale PV system near Orlando [148].

Ref. [149], a stochastic machine learning framework is developed to forecast outages in specific distribution networks, and the framework is shown to have transfer learning capability. In Ref. [150], a novel scheme is presented to achieve resilience of distribution networks in a time shared manner and by using relatively small distributed energy resources but adding a small number of circuit reclosers that enable segmentation and dynamic microgrids. And, PV + storage can further enhance the resilence solution as PV not only reduce local energy needs but also potentially charge the battery storage systems and in turn extend their operational periods [151].

# 5. Policy solutions

Individuals and communities play an important role in building community resilience. As distributed energy resources become more common, and consumers transition to prosumers that contribute to the design and supply of the electricity grid, the models for how resilient grids are governed and incentivized are changing [152,153]. Traditionally, electricity is provided through centralized electric utility providers connected to large power plants that generate electricity that is transmitted across the grid to consumers. However, smallscale PV systems (less than 1 MW in size) in the United States have grown by over 800 % since 2014 from 5000 MW to almost 48,000 MW by December 2023, dramatically increasing the amount of distributed resources on the grid [154].

This growth in small-scale PV in the US is attributed to lower PV module hardware and soft costs (e.g. installation, financing, etc.), as well as state and federal clean energy incentives and policies to reduce the cost of adding solar arrays [155,156]. One example is the federal residential solar investment tax credit (ITC), which provides solar PV owners a tax credit for up to 26 % of the cost of a new residential solar array in their federal income taxes, greatly reducing the installation cost of a new PV system (Table 2). The Inflation Reduction Act of 2022 raises this tax credit to 30 % for installations between 2022 and 2032 and is expected to continue promoting solar growth in the U.S [157].

Despite the recent growth in small-scale PV and cost reduction trends, studies show that small-scale solar is still in the early phase of demand, with only 2.2 % of homeowners across the United States with solar PV systems on-site [167]. Capital costs for installing distributed energy resources remain one of the top barriers to more wide-scale adoption. State adoption rates for rooftop solar range from 23.6 % in Hawaii and 11 % in California to almost no rooftop solar in states such as North and South Dakota and West Virginia [154]. While some of these variations are attributed to access to solar resources and electricity prices, state and local policies significantly impact the accessibility and affordability of distributed, customer-side solar [164,165,172].

In an effort to decarbonize the electric grid, some states have put into place renewable or clean energy targets, such as Renewable Portfolio Standards present in 30 states plus the District of Columbia. Renewable Portfolio Standards set targets for renewable energy contributions to the overall electricity grid mix in that state. The literature is mixed on the degree of effectiveness of Renewable Portfolio Standards on growing solar capacity in the U.S [163–166]. While these may not directly lead to compensation for rooftop solar homeowners [165], renewable energy targets have generally driven other state policies meant to encourage PV growth [164].

One such policy is net metering, which compensates PV owners for any excess electricity generated by their solar array that is exported to the grid, usually on a per-kilowatt-hour basis [167]. As of 2022, 33 states offer some form of net metering credits, however, the rate of compensation (typically retail rate or less), capacity caps, or other rules may impact their effectiveness [165,167]. Much of the literature agrees that net metering has had a substantial impact on distributed solar growth in the U.S. [164,165], perhaps as much as a doubling of small-scale PV capacity between 2008 and 2018 [167].

However, some electric utilities are seeking state regulations to slow the transition to more distributed generation sources [168], oftentimes by pursuing alternative or reduced forms of net metering [169]. The goal of the utility is to reduce the degree of compensation to the homeowner for distributed generation supplied back to the grid. Utilities and state regulatory agencies may also add additional and unnecessary equipment requirements for rooftop solar in interconnection regulations, citing safety or other reasons, which can add additional and unnecessary costs across the country [173].

In addition to generation-related policies, programs to reduce the installation, financing, and permitting soft costs for distributed solar have had mixed success in incentivizing solar adoption. Some solutions discussed by Ref. [156] to reduce small-scale solar PV soft costs include using quote platforms, which are third-party systems to collect and compare multiple PV quotes in a standardized, online format. Customer aggregation such as through microgrids or community solar projects typically reduces transaction costs through interconnection among many or one small-scale PV source. According to Ref. [20], particular design elements of community solar projects including the ownership model, rate design, and subscriber enrollment play an important role in the success of these programs in expanding solar adoption and localized access to clean energy benefits.

Another possible solution to incentivize the adoption of distributed energy resources is by promoting the benefits of grid resilience. Not only is distributed solar a cleaner energy resource with lower carbon dioxide and traditional pollutants, but microgrids and small-scale networks of distributed generation coupled with energy storage can also enhance

#### Table 2

Summary of policy incentives aimed at lowering the cost of PV systems.

Policy	Description	Level of Government	Primary Benefits	Primary Drawbacks
Investment Tax Credit	Tax credit for up to 26 % cost of new residential solar array	Federal	Reduces overall investment cost making it cost-competitive with other generation sources [158], extended recently in the Inflation Reduction Act 2022 and expected to continue improving cost competitiveness of solar [157,159]	Most clean energy tax credits have gone to higher-income households [160,161], scheduled sun setting of the policy creates some uncertainty [158]
Renewable Portfolio Standards	Sets targets for renewable energy contributions to the grid	State	More stringent RPS leads to higher renewable capacity growth [162,163], encourages other state policies that more directly incentivize PV growth [164]	Not a direct compensation for rooftop solar homeowners [165], impact of RPS smaller in longer time series, and smaller compared to federal policy impact [166]
Net Metering	Compensates residential solar homeowners for generation exported to the grid	State/Utility	Reduces the payback time for covering solar investment costs [164,165]	Limitations such as lower rates of compensation or caps reduce degree of effectiveness [165,167–169], paid for in part by low-income ratepayers without solar [170,171]

electrical grid resilience [174,175]. [174] use a discrete choice experiment in a survey of over 900 New Yorkers to determine that respondents would typically be willing to pay an additional \$14 per month of their electricity bill for a microgrid system with higher service levels and greater reliability. However, literature on resilient energy systems also considers lifecycle questions regarding the environmental sustainability of materials used in PV + storage microgrids. Batteries and other energy storage material production and disposal are under the attention of scientists and policymakers [88,176,177]. Despite this, there is good agreement in the literature that solar panels and energy storage can have a powerful impact on climate mitigation and adaptation [178].

Despite policies that are working to reduce the capital costs of cleaner and more resilient, small-scale energy systems, the financial system and debt-based approach towards energy transitions are preventing even larger-scale deployment of these technology solutions [170,179]. For socioeconomic and medically vulnerable populations, who particularly stand to benefit from better grid resilience with distributed solar, the high investment cost for the technology can make it more difficult to attain. Numerous studies show that rooftop solar adopters tend to have higher incomes [180,181], live in White-majority census tracts [182], and have higher levels of education [181]. However, declining costs of solar [181], and some energy justice programs [182] are beginning to improve more equitable solar adoption across demographics.

Solar plus storage microgrids present an opportunity for greater sustainability towards climate mitigation, as well as more resilient energy solutions able to withstand the consequence of the changing climate. Therefore, if policymakers could transform from the financial market regulations perspective, more private entities and citizens would be able to step in and foster the energy transition via renewable energies [179].Based on a review of the literature, support for pro-solar policies such as net-metering [167], community shared solar programs [20], and energy justice programs [26] are promising avenues for improving more equitable access to clean energy resources. In order to develop a robust, community-based energy resilience plan, significant engagement with multiple communities, technology, and local government partners and community members is needed to develop effective and fair programs and incentives. Co-production of knowledge using direct feedback from community members has the potential to produce more effective solutions to complex and wicked environmental problems [183] and energy justice challenges [26]. However, more research is needed on the specific policy designs and implementation strategies that work best to improve solar adoption inequities and access to the benefits of affordable clean energy technology [170,182].

## 6. Conclusion

Climate change is exacerbating the number and magnitude of extreme weather events that are affecting communities worldwide. Compounding risks are disrupting efforts communities are making toward disaster risk reduction, sustainable development, and energy security and transition. Because of financial and infrastructural limitations, underserved communities, especially in coastal areas, are disproportionately affected by disasters resulting from climaterelated natural hazards. Power outages often occur in areas where there is a high reliance on the 'traditional grid' and little or no penetration of renewable energies, including solar PV. Indeed, 'traditional' sources of energy have proven to be less resilient to the effects of today's extreme weather events characteristics and this is creating a compounding effect on underserved communities who already struggle because of socioeconomic disadvantages.

Our study was motivated by the need to provide an overview of both the literature and the scientific efforts aimed at developing solutions to grid resilience challenges due to a changing climate. Thus, power outage modeling does not always accounts for climate change-related impacts that may affect a location in the future. Based on our study, research on short-term power outage forecast models under current climate conditions is ongoing, but there is a lack of focus on long-term climateinduced changes in power outages. From our review, weather parameters in long-term weather-outage models in addition to hurricane and temperature data are needed. Similarly, the relevance of flooding overall, and surge along the coastal regions, are parameters that, given climate change and its impact on characteristics of storms, should be included as well. This review reveals the necessity to evaluate the impacts of the uncertainties of each weather variable in the long-term weather-outage projections in order to assess the long-term changes in weather-related power outages to better plan a more resilient power grid. This is particularly important for vulnerable underserved communities where energy transition and access to multiple sources of energy appears to be a pathway to strengthen their resilience to climateinduced extreme weather events.

Numerous technical solutions to improve grid resilience using cleaner energy sources exist, and research in this field is focused on the increased need for diversification, decentralization, and integration of major generation assets. However, technical challenges still remain regarding complementary renewable energy technologies such as energy storage, as well as the study of integration algorithms for intermittent renewable sources of electricity. Ongoing research is focusing on the development of machine learning and optimized algorithms that can better forecast power outages in a given community. These algorithms can be scalable and computationally efficient but must be kept generic in order to do so; additionally, technological solutions and related economic analyses depend by a variety of factors, including the physical location of the system, the power system topology, the physical attributes of the distribution network itself, and the generation/load distribution of it. Further, the cost of these technologies, necessary upgrades, and transmission infrastructure makes this more challenging, especially for socially vulnerable populations that traditionally are not firstadopters in cleaner and more resilient energy technology.

Despite the resilience-related benefits to underserved communities, promotion of renewable resources as an alternative to traditional sources of energy still faces resistance in the energy transition. For example, electricity providers are considered about the loss of profit to more distributed solar PV. Furthermore, there is little data availability, especially granular data, that would help identify populations more subject to power outages in order to focus policy efforts where a higher number of people and/or underserved communities are most affected by those outages. Last but not least, policymakers should focus on providing solutions that would enable the energy market to overcome the current debt-based approach in favor of different types of incentives. In order to implement effective policies that favor energy security in underserved communities in the aftermath of extreme weather events, policymakers should access data about power outages at a more granular level that is currently available.

There had been recent progress on promoting energy equity in the U. S. at the federal level within the timeframe this study was conducted (2022-2023). For example, the Executive Order on Advancing Racial Equity and Support for Underserved Communities Through the Federal Government (EO 13985, January 20, 2021), federal agencies that promote clean energy are pursuing more transdisciplinary initiatives to promote equitable solutions to climate change. The U.S. Department of Energy (DOE) was moving forward with a set of initiatives and research projects funding aimed at promoting renewable energy use in socially vulnerable communities. The U.S. Environmental Protection Agency (EPA) was focusing its efforts on equity making sure that communities and other relevant stakeholders are involved in the decision-making process. While these federal efforts are currently under review in early 2025, state and local governments may build off of these initiatives to continue their efforts at a more local scale.

#### **Funding sources**

This material is based upon work supported by the U.S. Department of Energy's Office of Energy Efficiency and Renewable Energy (EERE) under the Solar Energy Technologies Office Award Numbers DE-EE0010418, DEEE0007998, DE-EE0009028, DE-EE0009152, and DE-EE0009339.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

## Data availability

No data was used for the research described in the article.

#### References

- Seneviratne SI, Zhang X, Adnan M, Badi W, Dereczynski C, Di Luca A, Vicente-Serrano SM, Wehner M, Zhou B. 11 chapter 11: weather and climate extreme events in a changing climate. 2021.
- [2] Otto IM, Donges JF, Cremades R, Bhowmik A, Hewitt RJ, Lucht W, Rockström J, Allerberger F, McCaffrey M, Doe SS, et al. Social tipping dynamics for stabilizing earth's climate by 2050. Proc Natl Acad Sci USA 2020;117(5):2354–65.
- U.S. Billion-Dollar Weather and Climate Disasters. https://doi.org/10.25921/s tkw-7w73. https://www.ncei.noaa.gov/access/billions/.
- [4] Smith AB, Katz RW. US billion-dollar weather and climate disasters: data sources, trends, accuracy and biases. Nat Hazards 2013;67(2):387–410. https://doi.org/ 10.1007/s11069-013-0566-5. 202409-24.
- [5] Smith AB, Matthews JL. Quantifying uncertainty and variable sensitivity within the US billion-dollar weather and climate disaster cost estimates. Nat Hazards 2015;77(3):1829–51. https://doi.org/10.1007/s11069-015-1678-x. [Accessed 24 September 2024].
- [6] Masson-Delmotte V, Zhai P, Pirani A, Connors S, P'ean C, Berger S, Caud N, Chen Y, Goldfarb L, Gomis M, et al. Contribution of working group i to the sixth assessment report of the intergovernmental panel on climate change. Climate change; 2021.
- [7] Kwasinski A, Andrade F, Castro-Sitiriche MJ, O'Neill-Carrillo E. Hurricane maria effects on Puerto Rico electric power infrastructure. IEEE Power and Energy Technology Systems Journal 2019;6(1):85–94.
- [8] Core ML, Fisher E, Morgan JD, Ramachandran B, Vakiti S. Lights-out after hurricane michael: a spatially informed bayesian network analysis of power outages. SE Geogr 2022;62(2):128–46.
- [9] Zhang G, Zhong H, Tan Z, Cheng T, Xia Q, Kang C. Texas electric power crisis of 2021 warns of a new blackout mechanism. CSEE journal of Power and Energy Systems 2022;8(1):1–9.
- [10] Do V, McBrien H, Flores NM, Northrop AJ, Schlegelmilch J, Kiang MV, Casey JA. Spatiotemporal distribution of power outages with climate events and social vulnerability in the USA. Nat Commun 2023;14(1):2470.
- [11] Li'evanos RS, Horne C. Unequal resilience: the duration of electricity outages. Energy Pol 2017;108:201–11.
- [12] Coleman N, Esmalian A, Lee C-C, Gonzales E, Koirala P, Mostafavi A. Energy inequality in climate hazards: empirical evidence of social and spatial disparities in managed and hazard-induced power outages. Sustain Cities Soc 2023;92: 104491.
- [13] Emanuel K, Sundararajan R, Williams J. Hurricanes and global warming: results from downscaling ipcc ar4 simulations. Bull Am Meteorol Soc 2008;89(3): 347–68.
- [14] Becker P. Sustainability science: managing risk and resilience for sustainable development. ???: Elsevier; 2023.
- [15] Haimes YY. On the complex definition of risk: a systems-based approach. Risk Anal: Int J 2009;29(12):1647–54.
- [16] Charani Shandiz S, Foliente G, Rismanchi B, Wachtel A, Jeffers RF. Resilience framework and metrics for energy master planning of communities. Energy 2020; 203:117856. https://doi.org/10.1016/j.energy.2020.117856. [Accessed 4 October 2023].
- [17] Stout S, Lee N, Cox S, Elsworth J, Leisch J. Power sector resilience planning guidebook: a self-guided reference for practitioners. Resilient energy platform. Golden, CO: National Renewable Energy Laboratory (NREL), United States Agency for International Development (USAID); June 2019.
- U.S. Energy Information Administration State Energy Data System. https://www. eia.gov/state/seds/.
- [19] Google Project Sunroof. https://sunroof.withgoogle.com/.
- [20] Chan G, Evans I, Grimley M, Ihde B, Mazumder P. Design choices and equity implications of community shared solar. Electr J 2017;30(9):37-41.
- [21] Thind MPS, Tessum CW, Azevedo IL, Marshall JD. Fine particulate air pollution from electricity generation in the US: health impacts by race, income, and

#### Renewable and Sustainable Energy Reviews 213 (2025) 115434

geography. Environ Sci Technol 2019;53(23):14010–9. https://doi.org/10.1021/ acs.est.9b02527.Publisher:AmericanChemicalSociety. 2024-0927.

- [22] Issa A. Deaths related to hurricane irma—Florida, Georgia, and North Carolina, september 4–october 10, 2017. MMWR. Morbidity and Mortality Weekly Report 2018;67.
- [23] Valencia F, Billi M, Urquiza A. Overcoming energy poverty through micro-grids: an integrated framework for resilient, participatory sociotechnical transitions. Energy Res Social Sci 2021;75:102030.
- [24] O'Shaughnessy E, Barbose G, Wiser R, Forrester S, Darghouth N. The impact of policies and business models on income equity in rooftop solar adoption. Nat Energy 2021;6(1):84–91.
- [25] Meerow S, Pajouhesh P, Miller TR. Social equity in urban resilience planning. Local Environ 2019;24(9):793–808.
- [26] Carley S, Engle C, Konisky DM. An analysis of energy justice programs across the United States. Energy Pol 2021;152:112219.
- [27] Anadon LD, Chan G, Harley AG, Matus K, Moon S, Murthy SL, Clark WC. Making technological innovation work for sustainable development. Proc Natl Acad Sci USA 2016;113(35):9682–90.
- [28] Matin N, Forrester J, Ensor J. What is equitable resilience? World Dev 2018;109: 197–205.
- [29] McPhee C, Bliemel M, van der Bijl-Brouwer M. Transdisciplinary innovation. Technology innovation management review. 2018.
- [30] Akanbi OD, Bhuvanagiri DC, Barcelos EI, Nihar A, Gonzalez Hernandez B, Yarus JM, French RH. Integrating multiscale geospatial analysis for monitoring crop growth, nutrient distribution, and hydrological dynamics in large-scale agricultural systems. Journal of Geovisualization and Spatial Analysis 2024;8(1): 9.
- [31] Campbell RJ, Lowry S. Weather-related power outages and electric system resiliency. Washington, DC: Congressional Research Service, Library of Congress; 2012.
- [32] Henry D, Ramirez-Marquez JE. On the impacts of power outages during hurricane sandy—a resilience-based analysis. Syst Eng 2016;19(1):59–75.
- [33] Ghorbanzadeh M, Koloushani M, Ulak MB, Ozguven EE, Jouneghani RA. Statistical and spatial analysis of hurricane-induced roadway closures and power outages. Energies 2020;13(5):1098.
- [34] Doe U. Electric disturbance events (OE-417) annual summaries. 2000-2023.
- [35] Flores NM, McBrien H, Do V, Kiang MV, Schlegelmilch J, Casey JA. The 2021 Texas power crisis: distribution, duration, and disparities. J Expo Sci Environ Epidemiol 2023;33(1):21–31.
- [36] Doe U. Economic benefits of increasing electric grid resilience to weather outages. 2013.
- [37] Eia U. Form eia-861m (formerly eia-826) detailed data. 2017.
- [38] Casey JA, Fukurai M, Hern'andez D, Balsari S, Kiang MV. Power outages and community health: a narrative review. Current environmental health reports 2020;7:371–83.
- [39] Nejat A, Solitare L, Pettitt E, Mohsenian-Rad H. Equitable community resilience: the case of winter storm uri in Texas. Int J Disaster Risk Reduc 2022;77:103070.
- [40] Carvallo J, Hsu FC, Shah Z, Taneja J. Frozen out in Texas: blackouts and inequity. The Rockefeller Foundation; 2021.
- [41] Lee C-C, Maron M, Mostafavi A. Community-scale big data reveals disparate impacts of the Texas winter storm of 2021 and its managed power outage. Humanities and Social Sciences Communications 2022;9(1):1–12.
- [42] Brelsford C, Tennille S, Myers A, Chinthavali S, Tansakul V, Denman M, Coletti M, Grant J, Lee S, Allen K, Johnson E, Huihui J, Hamaker A, Newby S, Medlen K, Maguire D, Dunivan Stahl C, Moehl J, Redmon D, Sanyal J, Bhaduri B. A dataset of recorded electricity outages by United States county 2014–2022. Sci Data 2024;11(1):271. https://doi.org/10.1038/s41597-024-03095-5.Publisher: NaturePublishingGroup. 2024-05-28.
- [43] EAGLE-I Power Outage Data Information. https://doi.org/10.13139/ORNLNCC S/1975203. https://doi.ccs.ornl.gov/dataset/d26451af-7ea4-577c-8285-78b7 2cd8a2dc.
- [44] Stevens J, Rom'an M, Shrestha R. Lights out in michael's wake. https://earthobs ervatory.nasa.gov/images/92887/lights-out-in-michaels-wake; 2018.
- [45] Cerrai D, Wanik DW, Bhuiyan MAE, Zhang X, Yang J, Frediani ME, Anagnostou EN. Predicting storm outages through new representations of weather and vegetation. IEEE Access 2019;7:29639–54.
- [46] Cerrai D, Watson P, Anagnostou EN. Assessing the effects of a vegetation management standard on distribution grid outage rates. Elec Power Syst Res 2019;175:105909.
- [47] Wanik D, Anagnostou E, Hartman B, Frediani M, Astitha M. Storm outage modeling for an electric distribution network in northeastern USA. Nat Hazards 2015;79:1359–84.
- [48] Nateghi R, Guikema SD, Quiring SM. Comparison and validation of statistical methods for predicting power outage durations in the event of hurricanes. Risk Anal: Int J 2011;31(12):1897–906.
- [49] Panteli M, Mancarella P. Modeling and evaluating the resilience of critical electrical power infrastructure to extreme weather events. IEEE Syst J 2015;11 (3):1733–42.
- [50] Staid A, Guikema SD, Nateghi R, Quiring SM, Gao MZ. Simulation of tropical cyclone impacts to the us power system under climate change scenarios. Climatic Change 2014;127:535–46.
- [51] Yang F, Watson P, Koukoula M, Anagnostou EN. Enhancing weather-related power outage prediction by event severity classification. IEEE Access 2020;8: 60029–42.

#### S. Belligoni et al.

- [52] Mukherjee S, Nateghi R, Hastak M. A multi-hazard approach to assess severe weather-induced major power outage risks in the us. Reliab Eng Syst Saf 2018; 175:283–305.
- [53] Guikema SD, Nateghi R, Quiring SM, Staid A, Reilly AC, Gao M. Predicting hurricane power outages to support storm response planning. IEEE Access 2014; 2:1364–73.
- [54] Nateghi R, Guikema SD, Quiring SM. Forecasting hurricane induced power outage durations. Nat Hazards 2014;74:1795–811.
- [55] Maia-Silva D, Kumar R, Nateghi R. The critical role of humidity in modeling summer electricity demand across the United States. Nat Commun 2020;11(1): 1686.
- [56] Mukherjee S, Nateghi R. Climate sensitivity of end-use electricity consumption in the built environment: an application to the state of Florida, United States. Energy 2017;128:688–700.
- [57] Liu H, Davidson RA, Apanasovich TV. Statistical forecasting of electric power restoration times in hurricanes and ice storms. IEEE Trans Power Syst 2007;22(4): 2270–9.
- [58] Han S-R, Guikema SD, Quiring SM. Improving the predictive accuracy of hurricane power outage forecasts using generalized additive models. Risk Anal: Int J 2009;29(10):1443–53.
- [59] Yang F, Koukoula M, Emmanouil S, Cerrai D, Anagnostou EN. Assessing the power grid vulnerability to extreme weather events based on long-term atmospheric reanalysis. Stoch Environ Res Risk Assess 2023:1–16.
- [60] DOE, U.: US energy sector vulnerabilities to climate change. 2013.
- [61] DOE, U.: climate change and the U.S. Energy sector: regional vulnerabilities and resilience solutions. 2015.
- [62] Alemazkoor N, Rachunok B, Chavas DR, Staid A, Louhghalam A, Nateghi R, Tootkaboni M. Hurricane-induced power outage risk under climate change is primarily driven by the uncertainty in projections of future hurricane frequency. Sci Rep 2020;10(1):15270.
- [63] Emrich CT, Aksha SK, Zhou Y. Assessing distributive inequities in fema's disaster recovery assistance fund allocation. Int J Disaster Risk Reduc 2022;74:102855.
- [64] Cutter SL. Vulnerability to environmental hazards. Prog Hum Geogr 1996;20(4): 529–39.
- [65] Busby JW, Baker K, Bazilian MD, Gilbert AQ, Grubert E, Rai V, Rhodes JD, Shidore S, Smith CA, Webber ME. Cascading risks: understanding the 2021 winter blackout in Texas. Energy Res Social Sci 2021;77:102106.
- [66] Abedi A, Gaudard L, Romerio F. Review of major approaches to analyze vulnerability in power system. Reliab Eng Syst Saf 2019;183:153–72.
- [67] Cutter SL, Boruff BJ, Shirley WL. Social vulnerability to environmental hazards. Soc Sci Q 2003;84(2):242–61.
  [68] Wisner B. Vulnerability as concept, model, metric, and tool. Oxford University
- [68] Wisner B. Vulnerability as concept, model, metric, and tool. Oxford University Press; 2016. https://doi.org/10.1093/acrefore/9780199389407.013.25. https ://oxfordre.com/naturalhazardscience/view/10.1093/acrefore/978019938940 7.001.0001/acrefore-9780199389407-e-25.
- [69] Morath DP. Social vulnerability and public health: developing a metric for medical emergency management in Florida. University of South Carolina; 2010. PhD thesis.
- [70] Hames E, Stoler J, Emrich CT, Tewary S, Pandya N. A gis approach to identifying socially and medically vulnerable older adult populations in south Florida. Gerontol 2017;57(6):1133–41.
- [71] Emrich CT, Morath D, Buck K. Incorporating medical and social vulnerability into an all-hazards assessment for the state of Florida. Final Report to the Florida Department of Health; 2012.
- [72] Perrow C. The next catastrophe: reducing our vulnerabilities to natural, industrial, and terrorist disasters. ???: Princeton University Press; 2011.
- [73] Dugan J, Byles D, Mohagheghi S. Social vulnerability to longduration power outages. Int J Disaster Risk Reduc 2023;85:103501.
- [74] Boyle E, Inanlouganji A, Carvalhaes T, Jevti'c P, Pedrielli G, Reddy TA. Social vulnerability and power loss mitigation: a case study of Puerto Rico. Int J Disaster Risk Reduc 2022;82:103357.
- [75] Karakoc DB, Barker K, Zobel CW, Almoghathawi Y. Social vulnerability and equity perspectives on interdependent infrastructure network component importance. Sustain Cities Soc 2020;57:102072.
- [76] Lin S, Fletcher BA, Luo M, Chinery R, Hwang S-A. Health impact in new york city during the northeastern blackout of 2003. Publ Health Rep 2011;126(3):384–93.
- [77] Anderson GB, Bell ML. Lights out: impact of the august 2003 power outage on mortality in new york, ny. Epidemiology 2012;23(2):189.
- [78] Cruz-Cano R, Mead EL. Causes of excess deaths in Puerto Rico after hurricane maria: a time-series estimation. Am J Publ Health 2019;109(7):1050–2.
- [79] Molinari NAM, Chen B, Krishna N, Morris T. Who's at risk when the power goes out? the at-home electricity-dependent population in the United States, 2012. J Publ Health Manag Pract: JPHMP 2017;23(2):152.
- [80] Dominianni C, Ahmed M, Johnson S, Blum M, Ito K, Lane K. Power outage preparedness and concern among vulnerable new york city residents. J Urban Health 2018;95:716–26.
- [81] Bhattacharyya A, Hastak M. A data-driven approach to quantify disparities in power outages. Sci Rep 2023;13(1):7247.
- [82] Kearns RD, Wigal MS, Fernandez A, Tucker MA, Zuidgeest GR, Mills MR, Cairns BA, Cairns CB. The 2012 derecho: emergency medical services and hospital response. Prehospital Disaster Med 2014;29(5):542–5.
- [83] The resilience roadmap. www.nrel.gov/resilience-planningroadmap.
- [84] Jufri FH, Widiputra V, Jung J. State-of-the-art review on power grid resilience to extreme weather events: definitions, frameworks, quantitative assessment methodologies, and enhancement strategies. Appl Energy 2019;239:1049–65.

#### Renewable and Sustainable Energy Reviews 213 (2025) 115434

- [85] Panteli M, Mancarella P. Modeling and evaluating the resilience of critical electrical power infrastructure to extreme weather events. IEEE Syst J 2017;11 (3):1733–42. https://doi.org/10.1109/JSYST.2015.2389272.
- [86] Time-series hosting capacity analysis considering existing der. Report #0000000300201552. Electric Power Research Institute; 2021.
- [87] Multi-agent opendss. https://www.ece.ucf.edu/qu/ma-opendss/.
- [88] Xu Y, Sun W, Qu Z. Renewable energy integration and system operation challenge: control and optimization of millions of devices. New technologies for power system operation and analysis. 2020.
- [89] Wiser R, Millstein D, Mai T, MacKnick J, Carpenter A, Cohen S, Cole W, Frew B, Heath G. The environmental and public health benefits of achieving high penetrations of solar energy in the United States. Energy 2016;113:472–86. https://doi.org/10.1016/j.energy.2016.07.068. [Accessed 30 September 2024].
- [90] Prehoda EW, Pearce JM. Potential lives saved by replacing coal with solar photovoltaic electricity production in the U.S. Renew Sustain Energy Rev 2017; 80:710–5. https://doi.org/10.1016/j.rser.2017.05.119. [Accessed 30 September 2024].
- [91] Abel D, Holloway T, Harkey M, Rrushaj A, Brinkman G, Duran P, Janssen M, Denholm P. Potential air quality benefits from increased solar photovoltaic electricity generation in the Eastern United States. Atmos Environ 2018;175: 65–74. https://doi.org/10.1016/j.atmosenv.2017.11.049. [Accessed 30 September 2024].
- [92] Bolinger M, Seel J, Kemp JM, Warner C, Katta A, Robson D. Utility-scale solar, 2023 edition: empirical trends in deployment, technology, cost, performance, PPA pricing, and value in the United States. Technical report October 2023. htt ps://escholarship.org/uc/item/9m7260r5. 2024-10-02.
- [93] Barbose G, Darghouth N, O'Shaughnessy E, Forrester S. Tracking the sun. Technical report 2024. https://emp.lbl.gov/sites/default/files/2024-08/Tracking %20the%20Sun%202024Report.pdf, 2024-10-20.
- [94] Ropp ME, Begovic M, Rohatgi A. Prevention of islanding in gridconnected photovoltaic systems. Prog Photovoltaics Res Appl 1999;7(1):39–59. https://doi. org/10.1002/(SICI)1099-159X(199901/02)7:1(39::AID-PIP246)3.0.CO;2-J. 202410-20.
- [95] Woyte A, De Brabandere K, Van Dommelen D, Belmans R, Nijs J. International harmonization of grid connection guidelines: adequate requirements for the prevention of unintentional islanding. Prog Photovoltaics Res Appl 2003;11(6): 407–24. https://doi.org/10.1002/pip.503.eprint. [Accessed 20 October 2024]. https://onlinelibrary.wiley.com/doi/pdf/10.1002/pip.503.
- [96] Reedy R, Davis K, Click D, Ropp M, Shaffer A. Power line carrier permissive as a simple and safe method of enabling inverter ridethrough operation of distributed grid-tied photovoltaic systems. In: 2011 IEEE international symposium on power line communications and its applications; 2011. p. 209–12. https://doi.org/ 10.1109/ISPLC.2011.5764394. https://ieeexplore.ieee.org/abstract/documen t/5764394. 2024-10-20.
- [97] Baik S, Davis AL, Morgan MG. Assessing the cost of large-scale power outages to residential customers. Risk Anal 2018;38(2):283–96.
- [98] Kobayashi, T., Katayama, H., Kinden, Y., Kato, Y., Aya, Y., Hashiguchi, T., Kanematsu, D., Kobayashi, T., Terakawa, A., Fujiwara, H.: Cost-efficiency potential of solar energy on a global scale: Case studies for Si solar modules with PERC and heterojunction structures. Progress in Photovoltaics: Research and Applications n/a(n/a). https://doi.org/10.1002/pip.3835.eprint: https:// onlinelibrary.wiley.com/doi/pdf/10.1002/pip.3835. Accessed 2024-07-17.
- [99] Routhier AF, Bowden SG, Goodnick SM, Honsberg CB. What is the LCOE of residential solar + battery in the face on increasingly complex utility rate plans?. In: 2021 IEEE 48th photovoltaic specialists conference (PVSC); 2021. p. 2074–8. https://doi.org/10.1109/PVSC43889.2021.9519070.ISSN:0160-8371. https://ieeexplore.ieee.org/abstract/document/9519070 Accessed 202410-20.
- [100] Schneller EJ, Davis KO. 12 outdoor field testing: environmental and system stress factors. In: Yang HE, French RH, Bruckman LS, editors. Durability and reliability of polymers and other materials in photovoltaic modules. Plastics design library. William Andrew Publishing, ???; 2019. p. 279–95. https://doi. org/10.1016/B978-0-12-811545-9.00012-4. https://www.sciencedirect.com/sci ence/article/pii/B9780128115459000124. 2024-10-24.
- [101] Davis KO, Kurtz SR, Jordan DC, Wohlgemuth JH, SorloaicaHickman N. Multipronged analysis of degradation rates of photovoltaic modules and arrays deployed in Florida. Prog Photovoltaics Res Appl 2013;21(4):702–12. https://doi. org/10.1002/pip.2154. [Accessed 21 February 2018]. Number: 4.
- [102] Jordan DC, Sekulic B, Marion B, Kurtz SR. Performance and aging of a 20-year-old silicon PV system. IEEE J Photovoltaics 2015;5:744–51. https://doi.org/10.1109/ JPHOTOV.2015.2396360.
- [103] Dubey R, Chattopadhyay S, Kuthanazhi V, Kottantharayil A, Singh Solanki C, Arora BM, Narasimhan KL, Vasi J, Bora B, Singh YK, Sastry OS. Comprehensive study of performance degradation of field-mounted photovoltaic modules in India. Energy Sci Eng 2017;5(1):51–64. https://doi.org/10.1002/ese3.150. [Accessed 10 January 2020].
- [104] Annigoni E, Virtuani A, Levrat J, Faes A, Sculati-Meillaud F, Despeisse M, Ballif C. Quantifying and modeling the impact of interconnection failures on the electrical performance of crystalline silicon photovoltaic modules. Prog Photovoltaics Res Appl 2019;27(5):424–32. https://doi.org/10.1002/pip.3111. [Accessed 24 April 2019]. Number: 5.
- [105] Jordan DC, Marion B, Deline C, Barnes T, Bolinger M. PV field reliability status—analysis of 100 000 solar systems. Prog Photovoltaics Res Appl 2020;28 (8):739–54. https://doi.org/10.1002/pip.3262. [Accessed 18 March 2020]. http s://onlinelibrary.wiley.com/doi/pdf/10.1002/pip.3262.

- [106] Colvin DJ, Schneller EJ, Davis KO. Impact of interconnection failure on photovoltaic module performance. Prog Photovoltaics Res Appl 2021;29(5): 524–32. https://doi.org/10.1002/pip.3401. [Accessed 26 January 2022].
- [107] Kim J, Rabelo M, Padi SP, Yousuf H, Cho E-C, Yi J. A review of the degradation of photovoltaic modules for life expectancy. Energies 2021;14(14):4278. https:// doi.org/10.3390/en14144278.Number:14Publisher: MultidisciplinaryDigitalPublishingInstitute. [Accessed 20 October 2024].
- [108] Kumar S, Meena R, Gupta R. Finger and interconnect degradations in crystalline silicon photovoltaic modules: a review. Sol Energy Mater Sol Cell 2021;230: 111296. https://doi.org/10.1016/j.solmat.2021.111296. [Accessed 2 September 2021].
- [109] Aghaei M, Fairbrother A, Gok A, Ahmad S, Kazim S, Lobato K, Oreski G, Reinders A, Schmitz J, Theelen M, Yilmaz P, Kettle J. Review of degradation and failure phenomena in photovoltaic modules. Renew Sustain Energy Rev 2022; 159:112160. https://doi.org/10.1016/j.rser.2022.112160. [Accessed 8 December 2022].
- [110] Colvin DJ, Davis KO. Degradation of edge-defined film-fed silicon glass–glass modules on Florida rooftop after 22 years. Phys Status Solidi Rapid Res Lett 2022; 16(11):2200215. https://doi.org/10.1002/pssr.202200215.eprint. [Accessed 1 October 2024]. https://onlinelibrary.wiley.com/doi/pdf/10.1002/pssr. 202200215.
- [111] Curran AJ, Yu X, Liu J, Colvin DJ, Iqbal N, Moran T, Brownell B, Li M, Davis KO, Huey BD, Jaubert J-N, Braid JL, Bruckman LS, French RH. Field studies of PERC and Al-BSF PV module performance loss using power and I-V timeseries. Front Energy Res 2023;11. [Accessed 5 February 2024].
- [112] Jordan DC, Deline C, Johnston S, Rummel SR, Sekulic B, Hacke P, Kurtz SR, Davis KO, Schneller EJ, Sun X, Alam MA, Sinton RA. Silicon heterojunction system field performance. IEEE J Photovoltaics 2018;8(1):177–82. https://doi. org/10.1109/JPHOTOV.2017.2765680. [Accessed 17 January 2022].
- [113] Karas J, Sinha A, Buddha VSP, Li F, Moghadam F, TamizhMani G, Bowden S, Augusto A. Damp heat induced degradation of silicon heterojunction solar cells with Cu-plated contacts. IEEE J Photovoltaics 2020;10(1):153–8. https://doi.org/ 10.1109/JPHOTOV.2019.2941693. 2021-10-04.
- [114] Liu W, Zhang L, Yang X, Shi J, Yan L, Xu L, Wu Z, Chen R, Peng J, Kang J, Wang K, Meng F, De Wolf S, Liu Z. Damp-heat-stable, high-efficiency, industrial-size silicon heterojunction solar cells. Joule 2020;4(4):913–27. https://doi.org/ 10.1016/j.joule.2020.03.003. [Accessed 28 November 2022].
- [115] Sulas-Kern DB, Owen-Bellini M, Ndione P, Spinella L, Sinha A, Ulicn'a S, Johnston S, Schelhas LT. Electrochemical degradation modes in bifacial silicon photovoltaic modules. Prog Photovoltaics Res Appl 2022;30(8):948–58. https:// doi.org/10.1002/pip.3530.eprint. [Accessed 4 October 2023]. https://onlinelib rary.wiley.com/doi/pdf/10.1002/pip.3530.
- [116] Iqbal N, Li M, Sakthivel TS, Mikeska K, Lu M, Nandakumar N, Duttagupta S, Dhamrin M, Tsuji K, Bowden S, Augusto A, Guan Y, Seal S, Davis KO. Impact of acetic acid exposure on metal contact degradation of different crystalline silicon solar cell technologies. Sol Energy Mater Sol Cell 2023;250:112089. https://doi. org/10.1016/j.solmat.2022.112089. [Accessed 4 January 2024].
- [117] Johnston SW, Jordan DC, Kern DB, Colvin DJ, Davis KO, Moutinho HR, Kroeger GF. Degradation-related defect level in weathered silicon heterojunction modules characterized by deep level transient spectroscopy. Sol Energy Mater Sol Cell 2023;262:112527. https://doi.org/10.1016/j.solmat.2023.112527. [Accessed 21 October 2024].
- [118] Sen C, Wang H, Wu X, Khan MU, Chan C, Abbott M, Hoex B. Four failure modes in silicon heterojunction glass-backsheet modules. Sol Energy Mater Sol Cell 2023; 257:112358. https://doi.org/10.1016/j.solmat.2023.112358. [Accessed 27 June 2023].
- [119] Sommeling PM, Liu J, Kroon JM. Corrosion effects in bifacial crystalline silicon PV modules; interactions between metallization and encapsulation. Sol Energy Mater Sol Cell 2023;256:112321. https://doi.org/10.1016/j. solmat.2023.112321. [Accessed 28 May 2024].
- [120] Theristis M, Stein JS, Deline C, Jordan D, Robinson C, Sekulic W, Anderberg A, Colvin DJ, Walters J, Seigneur H, King BH. Onymous early-life performance degradation analysis of recent photovoltaic module technologies. Prog Photovoltaics Res Appl 2023;31(2):149–60. https://doi.org/10.1002/pip.3615. eprint. [Accessed 20 October 2024]. https://onlinelibrary.wiley.com/doi/pdf/ 10.1002/pip.3615.
- [121] Nalin Venkat S, Yu X, Liu J, Wegmueller J, Jimenez JC, Barcelos EI, Aung HH, Li X, Jaubert J-N, French RH, Bruckman LS. Statistical analysis and degradation pathway modeling of photovoltaic minimodules with varied packaging strategies. Front Energy Res 2023;11. https://doi.org/10.3389/fenrg.2023.1127796. Publisher:Frontiers. [Accessed 28 May 2024].
- [122] Sinha A, Qian J, Moffitt SL, Hurst K, Terwilliger K, Miller DC, Schelhas LT, Hacke P. UV-induced degradation of high-efficiency silicon PV modules with different cell architectures. Prog Photovoltaics Res Appl 2023;31(1):36–51. https://doi.org/10.1002/pip.3606.eprint. [Accessed 3 October 2023]. https:// onlinelibrary.wiley.com/doi/pdf/10.1002/pip.3606.
- [123] Sen C, Wang H, Khan MU, Fu J, Wu X, Wang X, Hoex B. Buyer aware: three new failure modes in TOPCon modules absent from PERC technology. Sol Energy Mater Sol Cell 2024;272:112877. https://doi.org/10.1016/j. solmat.2024.112877. [Accessed 28 May 2024].
- [124] Jordan DC, Perry K, White R, Deline C. Extreme weather and PV performance. IEEE J Photovoltaics 2023;13(6):830–5. https://doi.org/10.1109/ JPHOTOV.2023.3304357. [Accessed 19 August 2024]. Conference Name: IEEE Journal of Photovoltaics.
- [125] Jackson ND, Gunda T. Evaluation of extreme weather impacts on utility-scale photovoltaic plant performance in the United States. Appl Energy 2021;302:

117508. https://doi.org/10.1016/j.apenergy.2021.117508. [Accessed 22 April 2022].

- [126] Johnston S, Sulas-Kern DB, Jordan D. Photovoltaic module imaging for hail damage assessment with multi-year follow up. In: 2021 IEEE 48th photovoltaic specialists conference (PVSC); 2021. p. 763–6. https://doi.org/10.1109/ PVSC43889.2021.9518491.ISSN:0160-8371. https://ieeexplore.ieee.org/abstra ct/document/9518491. 2024-08-19.
- [127] Gabor AM, Knodle PJ, Covino M, Fioresi J, Colvin DJ, Davis KO. Electroluminescence analysis and grading of hail damaged solar panels. IEEE J Photovoltaics 2022:1–6. https://doi.org/10.1109/JPHOTOV.2022.3206906. [Accessed 28 November 2022].
- [128] Seigneur H, Mohajeri N, Brooker RP, Davis KO, Schneller EJ, Dhere NG, Rodgers MP, Wohlgemuth J, Shiradkar NS, Scardera G, Rudack AC, Schoenfeld WV. Manufacturing metrology for c-Si photovoltaic module reliability and durability, Part I: feedstock, crystallization and wafering. Renew Sustain Energy Rev 2016;59:84–106. https://doi.org/10.1016/j.rser.2015.12.343. [Accessed 5 June 2024].
- [129] Schneller EJ, Brooker RP, Shiradkar NS, Rodgers MP, Dhere NG, Davis KO, Seigneur HP, Mohajeri N, Wohlgemuth J, Scardera G, Rudack AC, Schoenfeld WV. Manufacturing metrology for c-Si module reliability and durability Part III: module manufacturing. Renew Sustain Energy Rev 2016;59:992–1016. https:// doi.org/10.1016/j.rser.2015.12.215.
- [130] Davis KO, Rodgers MP, Scardera G, Brooker RP, Seigneur H, Mohajeri N, Dhere NG, Wohlgemuth J, Schneller E, Shiradkar N, Rudack AC, Schoenfeld WV. Manufacturing metrology for c-Si module reliability and durability Part II: cell manufacturing. Renew Sustain Energy Rev 2016;59:225–52. https://doi.org/ 10.1016/j.rser.2015.12.217.
- [131] Buerhop C, Fecher FW, Pickel T, Häring A, Adamski T, Camus C, Hauch J, Brabec CJ. Verifying defective PV-modules by IRimaging and controlling with module optimizers. Prog Photovoltaics Res Appl 2018;26(8):622–30. Error! Hyperlink reference not valid. [Accessed 24 July 2018].
- [132] Teubner J, Buerhop C, Pickel T, Hauch J, Camus C, Brabec CJ. Quantitative assessment of the power loss of silicon PV modules by IR thermography and its dependence on data-filtering criteria. Prog Photovoltaics Res Appl 2019;27(10): 856–68. https://doi.org/10.1002/pip.3175. Number: 0, [Accessed 23 September 2019].
- [133] Köntges M, Morlier A, Eder G, Fleiß E, Kubicek B, Lin J. Review: ultraviolet fluorescence as assessment tool for photovoltaic modules. IEEE J Photovoltaics 2020;10(2):616–33. https://doi.org/10.1109/JPHOTOV.2019.2961781.Number: 2ConferenceName:IEEEJournalofPhotovoltaics.
- [134] Li F, Buddha VSP, Schneller EJ, Iqbal N, Colvin DJ, Davis KO, TamizhMani G. Correlation of UV fluorescence images with performance loss of field-retrieved photovoltaic modules. IEEE J Photovoltaics 2021;11(4):926–35. https://doi.org/ 10.1109/JPHOTOV.2021.3075865. Conference Name: IEEE Journal of Photovoltaics, [Accessed 1 October 2024].
- [135] Chen X, Karin T, Jain A. Automated defect identification in electroluminescence images of solar modules. Sol Energy 2022;242:20–9. https://doi.org/10.1016/j. solener.2022.06.031. [Accessed 31 May 2024].
- [136] Fioresi J, Colvin DJ, Frota R, Gupta R, Li M, Seigneur HP, Vyas S, Oliveira S, Shah M, Davis KO. Automated defect detection and localization in photovoltaic cells using semantic segmentation of electroluminescence images. IEEE J Photovoltaics 2022;12(1):53–61. https://doi.org/10.1109/ JPHOTOV.2021.3131059. Conference Name: IEEE Journal of Photovoltaics.
- [137] Libby C, Paudyal B, Chen X, Hobbs WB, Fregosi D, Jain A. Analysis of PV module power loss and cell crack effects due to accelerated aging tests and field exposure. IEEE J Photovoltaics 2023;13(1):165–73. https://doi.org/10.1109/ JPHOTOV.2022.3228104. Conference Name: IEEE Journal of Photovoltaics.
- [138] Colvin D, Gabor AM, Oltjen WC, Knodle PJ, Yao AD, Thompson BA, Khan N, Lotfian S, Raby J, Jojo A, Yu X, Liggett M, Seigneur HP, French RH, Bruckman LS, Li M, Davis KO. Ultraviolet fluorescence imaging for photovoltaic module metrology: best practices and survey of features observed in fielded modules. TechRxiv 2024. https://doi.org/10.36227/techrxiv.171440962.23003014/v1. [Accessed 1 October 2024].
- [139] Li F, Colvin DJ, Pavan Buddha VS, Davis KO, Tamizhmani G. Electroluminescence and infrared imaging of fielded photovoltaic modules: a complementary analysis of series resistance-related defects. Sol Energy 2024;276:112704. https://doi.org/ 10.1016/j.solener.2024.112704. [Accessed 1 October 2024].
- [140] Stein JS, Holmgren WF, Forbess J, Hansen CW. PVLIB: open source photovoltaic performance modeling functions for Matlab and Python. In: 2016 IEEE 43rd photovoltaic specialists conference (PVSC); 2016. p. 3425–30. https://doi.org/ 10.1109/PVSC.2016.7750303. https://ieeexplore.ieee.org/document/775 0303/?arnumber=7750303. 2024-10-20.
- [141] Holmgren WF, Hansen CW, Mikofski MA. Pvlib python: a python package for modeling solar energy systems. J Open Source Softw 2018;3(29):884. https://doi. org/10.21105/joss.00884. [Accessed 9 June 2020]. Number: 29.
- [142] Jensen AR, Anderson KS, Holmgren WF, Mikofski MA, Hansen CW, Boeman LJ, Loonen R. Pvlib iotools—open-source Python functions for seamless access to solar irradiance data. Sol Energy 2023;266:112092. https://doi.org/10.1016/j. solener.2023.112092. [Accessed 24 October 2024].
- [143] System advisor Model<sup>™</sup> version 2023.12.17 (SAM<sup>™</sup> 2023.12.17). National Renewable Energy Laboratories; 2024. https://sam.nrel.gov.
- [144] PySAM. National renewable energy laboratories. https://github.com/nrel/pysam; 2024.
- [145] PVsyst. PVsyst SA. https://www.pvsyst.com/.
- [146] Karimi AM, Wu Y, Koyuturk M, French RH. Spatiotemporal graph neural network for performance prediction of photovoltaic power systems. Proc AAAI Conf Artif

#### S. Belligoni et al.

Intell 2021;35(17):15323–30. https://doi.org/10.1609/aaai.v35i17.17799. [Accessed 20 October 2024]. Number: 17.

- [147] Yu J, Wang Z, Majumdar A, Rajagopal R. Deepsolar: a machine learning framework to efficiently construct a solar deployment database in the United States. Joule 2018;2(12):2605–17.
- [148] Fujita, K.S., Ancona, Z.H., Kramer, L.A., Straka, M., Gautreau, T.E., Garrity, C.P., Robson, D., Diffendorfer, J.E., Hoen, B.: United States Large-Scale Solar Photovoltaic Database. https://doi.org/10.5066/P9IA3TUS. https://doi.org/ 10.5066/P9IA3TUS.
- [149] McDonald, K., Le, C., Qu, Z.: Open-source modeling of extreme weather impact on distribution networks. In: 3rd international workshop on open source modelling and simulation of energy systems, Vienna, Austria (September 3-6, 2024).
- [150] Tu, B., McDonald, K., Qu, Z., Simaan, M.A., Davis, K., Stevens, K.: Community based resilience of distribution systems against extreme weather: a time-sharing framework. In: IEEE international conference on green energy and smart systems, long beach, CA (November 4-5, 2024).
- [151] Tu B, Qu Z. Roles of PV+Storage at community centers in achieving a level of distribution network resilience after extreme weather conditions. In: 13th international conference on renewable energy research and applications. Japan: Nagasaki; November .
- [152] Perger T, Wachter L, Fleischhacker A, Auer H. Pv sharing in local communities: peer-to-peer trading under consideration of the prosumers' willingness-to-pay. Sustain Cities Soc 2021;66:102634.
- [153] Knodt M, Stöckl A, Steinke F, Pietsch M, Hornung G, Stroscher J-P. Power blackout: citizens' contribution to strengthen urban resilience. Energy Pol 2023; 174:113433.
- [154] Eia U. Form eia-861m (formerly eia-826) detailed data. 2023.
- [155] Barbose G, Darghouth NR, Weaver S, Feldman D, Margolis R, Wiser R. Tracking us photovoltaic system prices 1998–2012: a rapidly changing market. Prog Photovoltaics Res Appl 2015;23(6):692–704.
- [156] O'Shaughnessy E, Nemet GF, Pless J, Margolis R. Addressing the soft cost challenge in us small-scale solar pv system pricing. Energy Pol 2019;134:110956.
- [157] Larsen J, King B, Kolus H, Dasari N, Hiltbrand G, Herndon W. A turning point for us climate progress: assessing the climate and clean energy provisions in the inflation reduction act. Rhodium Group 2022;12.
- [158] Comello S, Reichelstein S. The us investment tax credit for solar energy: alternatives to the anticipated 2017 step-down. Renew Sustain Energy Rev 2016; 55:591–602.
- [159] Bistline JE, Mehrotra NR, Wolfram C. Economic implications of the climate provisions of the inflation reduction act. Brookings Pap Econ Activ 2023;2023(1): 77–182.
- [160] Borenstein S, Davis LW. The distributional effects of us clean energy tax credits. Tax Pol Econ 2016;30(1):191–234.
- [161] Borenstein S, Davis LW. The distributional effects of us tax credits for heat pumps, solar panels, and electric vehicles. Technical report, National Bureau of Economic Research 2024.
- [162] Yin H, Powers N. Do state renewable portfolio standards promote in-state renewable generation. Energy Pol 2010;38(2):1140–9.

#### Renewable and Sustainable Energy Reviews 213 (2025) 115434

- [163] Shrimali G, Chan G, Jenner S, Groba F, Indvik J, et al. Evaluating renewable portfolio standards for in-state renewable deployment: accounting for policy heterogeneity. Economics of Energy & Environmental Policy 2015;4(2):127–42.
- [164] Michaud G, Pitt D. Non-utility photovoltaic deployment: evaluation of us statelevel policy drivers. Strat Plann Energy Environ 2019;38(3):52–80.
- [165] Lemay AC, Wagner S, Rand BP. Current status and future potential of rooftop solar adoption in the United States. Energy Pol 2023;177:113571.
- [166] Mullen JD, Dong L. Effects of state and federal policy on renewable electricity generation capacity in the United States. Energy Econ 2022;105:105764.
  [167] Ros AJ, Sai SS. Residential rooftop solar demand in the us and the impact of net
- energy metering and electricity prices. Energy Econ 2023;118:106491.
   [168] Stokes LC. Short circuiting policy: interest groups and the battle over clean energy
- and climate policy in the American states. USA: Oxford University Press; 2020. [169] Hess DJ. The politics of niche-regime conflicts: distributed solar energy in the
- United States. Environ Innov Soc Transit 2016;19:42–50.
  [170] Brown MA, Soni A, Lapsa MV, Southworth K, Cox M. High energy burden and low-income energy affordability: conclusions from a literature review. Progress in Energy 2020;2(4):042003.
- [171] Carley S, Evans TP, Graff M, Konisky DM. A framework for evaluating geographic disparities in energy transition vulnerability. Nat Energy 2018;3(8):621–7.
- [172] Carley S. Distributed generation: an empirical analysis of primary motivators. Energy Pol 2009;37(5):1648–59.
- [173] Stevens KA, Iman S, Davis KO. The cost of utility discretion on residential solar requirements. Renew Sustain Energy Rev 2022;159:112231.
- [174] Hotaling C, Bird S, Heintzelman MD. Willingness to pay for microgrids to enhance community resilience. Energy Pol 2021;154:112248.
- [175] Gonz'alez DML, Rendon JG. Opportunities and challenges of mainstreaming distributed energy resources towards the transition to more efficient and resilient energy markets. Renew Sustain Energy Rev 2022;157:112018.
- [176] Larcher D, Tarascon J-M. Towards greener and more sustainable batteries for electrical energy storage. Nat Chem 2015;7(1):19–29.
- [177] Chowdhury MS, Rahman KS, Chowdhury T, Nuthammachot N, Techato K, Akhtaruzzaman M, Tiong SK, Sopian K, Amin N. An overview of solar photovoltaic panels' end-of-life material recycling. Energy Strategy Rev 2020;27: 100431.
- [178] Asutosh A, Woo J, Kouhirostami M, Sam M, Khantawang A, Cuales C, Ryor W, Kibert C. Renewable energy industry trends and its contributions to the development of energy resilience in an era of accelerating climate change. Int J Energy Power Eng 2020;14(8):233–40.
- [179] Polzin F. Mobilizing private finance for low-carbon innovation-a systematic review of barriers and solutions. Renew Sustain Energy Rev 2017;77:525–35.
- [180] Scheier E, Kittner N. A measurement strategy to address disparities across household energy burdens. Nat Commun 2022;13(1):288.
- [181] Darghouth NR, O'Shaughnessy E, Forrester S, Barbose G. Characterizing local rooftop solar adoption inequity in the us. Environ Res Lett 2022;17(3):034028.
- [182] Gao X, Zhou S. Solar adoption inequality in the us: trend, magnitude, and solar justice policies. Energy Pol 2022;169:113163.
- [183] Rycroft-Malone J, McCormack B, Hutchinson AM, DeCorby K, Bucknall TK, Kent B, Schultz A, Snelgrove-Clarke E, Stetler CB, Titler M, et al. Realist synthesis: illustrating the method for implementation research. Implement Sci 2012;7:1–10.